Author's response to reviews

Title: A case of non-small cell carcinoma lung with choroidal metastasis as presenting sign, showing good response to chemotherapy alone.

Authors:

Abhishek Singh (hivneg@gmail.com)
Parul Singh (parulophtha@gmail.com)
Kamal Sahni (sahanik@lycos.com)
Preety Shukla (preety03@rediffmail.com)
Vikas Shukla (vikasshukla_08@rediffmail.com)
Nirdosh K Pant (s_nishant_2050@yahoo.com)

Version: 5 Date: 2 February 2010

Author's response to reviews:

Major revisions (We require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system. We note that you have updated the title in the manuscript but not in the submission system. Please amend this so that both are identical.

Changes in submission system done.

Ethnicity and nationality: Please include the ethnicity and nationality of the patient within the Case Presentation.

Ethnicity and nationality included.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Author list: In the manuscript, please amend the capitalisation for the author Vikas Shukla, and in the submission system please amend the capitalisation for the author Nirdosh Kumar Pant. Both require the first letter of the surname to be capitalised. The preferred style is “First name middle Initial Last name” (e.g. Joe F Bloggs).

Amendments done.

Underlining: Please remove all underlining from the manuscript. This includes the section headings in the abstract.

Underlinings removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
Manuscript checked.